**Author’s response to reviews**

**Title:** Participatory Learning and Action to address type 2 diabetes in rural Bangladesh: a qualitative process evaluation

**Authors:**

Joanna Morrison (joanna.morrison@ucl.ac.uk)

Kohenour Akter (mary05kh@gmail.com)

Hannah Jennings (hannah.jennings.11@ucl.ac.uk)

Tasmin Nahar (tasminpcp@dab-bd.org)

Abdul Kuddus (kudduspcp@dab-bd.org)

Sanjit Shaha (sanjitpcp@dab-bd.org)

Naveed Ahmed (naveedahmed225@gmail.com)

Carina King (c.king@ucl.ac.uk)

Hassan Haghparast-Bidgoli (h.haghparast-bidgoli@ucl.ac.uk)

Anthony Costello (cihdcostello@gmail.com)

A.K. Khan (president@dab-bd.org)

Kishwar Azad (kishwar.azad@googlemail.com)

Edward Fottrell (e.fottrell@ucl.ac.uk)

**Version:** 3  **Date:** 23 Sep 2019

**Author’s response to reviews:**

1. Full name of institutional review board

   In the Ethics approval and consent to participate statement of the Declarations, please provide the full name of the BADAS Ethical Review Committee. Please also add this abbreviation to the Abbreviations section of the Declarations.

   We have added the abbreviation and full name to the text.

2. Missing authors – CK, KA.
We note that the contributions of author CK and KA are missing from the Authors’ contributions statement of the Declarations. The individual contributions of ALL authors to the manuscript should be specified in this section. Guidance and criteria for authorship can be found in the online journal submission guidelines.

We have corrected the authorship declaration.

3. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have uploaded a clean manuscript.